Author’s response to reviews

Title: Brief Engagement and Acceptance Coaching for Community and Hospice Settings (the BEACHeS Study): Protocol for the development and pilot testing of an evidence-based psychological intervention to enhance wellbeing and aid transition into palliative care

Authors:

Nicholas Hulbert-Williams (n.hulbertwilliams@chester.ac.uk)
Sabrina Norwood (s.norwood@chester.ac.uk)
David Gillanders (david.gillanders@ed.ac.uk)
Anne Finucane (anne.finucane@mariecurie.org.uk)
Juliet Spiller (juliet.spiller@mariecurie.org.uk)
Jenny Strachan (jenny.strachan@mariecurie.org.uk)
Sue Millington (sue.millington@btopenworld.com)
Brooke Swash (b.swash@chester.ac.uk)

Version: 2 Date: 22 Jul 2019

Author’s response to reviews:

Please see the 'Table of Revisions' document attached with our resubmission.